Author's response to reviews

Title: Supplier-induced demand for psychiatric admissions in Northern New England

Authors:
Bradley V Watts (bradley.v.watts@dartmouth.edu)
Brian Shiner (brian.r.shiner@dartmouth.edu)
Gunnar Klauss (gklauss@wfubmc.edu)
William B Weeks (william.b.weeks@dartmouth.edu)

Version: 4 Date: 3 August 2011

Author's response to reviews:

Dear Dr. Alam,

Thank you for provisionally accepting our paper to BMC psychiatry (MS: 1634878393498515). All changes have been made as requested below.

Thank you,

Brian Shiner MD, MPH

Author list: Bradley V Watts, Brian Shiner, Gunnar Klauss and William B Weeks

Title: Supplier-induced demand for psychiatric admissions in Northern New England

Journal: BMC Psychiatry

MS ID: 1634878393498515

Dear Prof Shiner,

We are writing to request the following changes to the format of your manuscript, so that the published article conforms to journal style. The deadline for making these changes is Tuesday, 09 August 2011.

We may have edited the manuscript for you, please ensure that all your revisions are made to the most recent version of your manuscript, which can be accessed at the following link:

http://www.biomedcentral.com/media/man/1634878393498515/msd/main/0

Figures can be accessed here:

http://www.biomedcentral.com/media/man/1634878393498515/msd/FIG/1
http://www.biomedcentral.com/media/man/1634878393498515/msd/FIG/2

Please note that this is the proofing stage. All revised manuscripts submitted to BMC Psychiatry are regarded as being ready for publication. After publication,
you will not be able to make any corrections to errors present in this version. Please check your manuscript carefully for any typing or content errors, as these will be included in the published version. Please resubmit your manuscript here via the online submission system:
http://www.biomedcentral.com/author/manuscript/details/view.do?manuscriptId=1634878393498515

Formatting changes to be made:

-----------------------------

Major revisions (we require the author to make these changes)

Figures: You have cited a figure 3 on page 14 but there is no such figure uploaded. Please upload the figure or remove the citation.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Title page: There should be 3 separate sections for the author list, affiliations (indicated via superscript numbers) and email addresses. Please remember to include the country in the affiliation details.

Author list: Please remove all titles from the author list.

Title page: Please remove all manuscript data from the title page (word counts etc).

Tracking: Please remove all tracking and coloured text from the manuscript.

Tables: Please place the titles above the tables and the legends below. Please remove all vertical lines from table 2.

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Thank you

We have addressed all requested changes to the proofs as outlined below.